
Mauna Loa Solar Observatory Observer's Log _____ Fri Feb 6 17:32:26 GMT 2009 Year: 09 Doy: 037 Observer: koon WEATHER COMMENT: Fri Feb 6 17:32:27 GMT 2009 Thick altocumulus and cirrostratus overcast, wind=5mph from the SSE, temp=38F end COMMENT: Fri Feb 6 17:32:59 GMT 2009 No obs until clouds thin out. end WEATHER COMMENT: Fri Feb 6 22:39:33 GMT 2009 Still cloudy. end **MKIV PROBLEM**: Fri Feb 6 22:39:42 GMT 2009 I received the new camera at home yesterday. Today I tested the camera electrcally and it looks good, the Power Supply also doesn't have a problem providing power to it like the old camera problem where the voltage was pulled way down due to a short or something. I'm currently devising a plan to allow me to complete Step #1 of Joan replacement plan writeup. The camera and the beamsplitter and the FLC are all tightly placed and the mounting is still uncertain and possibly cross-connected in a complex manner. I'm looking for assembly drawings for those assemblies and Rich is searching for some in Boulder. I can proceed without drawings but would prefer to look at drawings first. The height mask for step #7 is in the top middle drawer of the roll-around tool chest upstairs. end **MKIV PROBLEM**: Fri Feb 6 23:25:51 GMT 2009 Will continue with the rest of Step #1 on Sunday. end Sat Feb 7 00:18:15 GMT 2009 MkIV